Author's response to reviews

Title: The effect of long term combined yoga practice on the basal metabolic rate of healthy Adults

Authors:

M S Chaya (chayapu@hotmail.com)
Kurpad A V (a.kurpad@iphor.res.in)
Nagendra H R (hrnagendra@rediffmail.com)
Nagarathna R (rn44@rediffmail.com)

Version: 2 Date: 20 June 2006

Author's response to reviews:

Dear Editor,

There was error in the title, which stands corrected now.

The correct title is "The effect of long term combined yoga practice on the basal metabolic rate of healthy Adults"

The "Adults" was inadvertently omitted in the title.

We regret the error.

Chaya